Author’s response to reviews

Title: The impact of a psychiatry clinical rotation on the attitude of South African final year medical students towards mental illness

Authors:

Caro De Witt (carodewitt@gmail.com)

Inge Smit (ingesmit@sun.ac.za)

Esme Jordaan (esme.jordaan@mrc.ac.za)

Liezl Koen (liezlk@sun.ac.za)

Dana Niehaus (djhn@sun.ac.za)

Ulla Alexandra Botha (ulla@sun.ac.za)

Version: 2 Date: 28 Mar 2019

Author’s response to reviews:

Cover letter and response to reviewers was uploaded as separate document under supplementary material.